# Impact of Antarctic meltwater forcing on East Asian climate under greenhouse warming

Jong-Seong Kug<sup>1</sup>, Ji-hoon Oh<sup>1</sup>, Wonsun Park<sup>2</sup>, Hyung-Gyu Lim<sup>3</sup>, Kyung Min Noh<sup>1</sup>, and Kyung Jin<sup>4</sup>

<sup>1</sup>POSTECH <sup>2</sup>GEOMAR Helmholtz Centre for Ocean Research Kiel <sup>3</sup>Pohang University of Science and Technology <sup>4</sup>KOPRI

November 23, 2022

#### Abstract

In recent decades, Antarctic ice-sheet/shelf melting has been accelerated, releasing freshwater into the Southern Ocean. It has been suggested that the meltwater flux could lead to cooling in the Southern Hemisphere, which would retard global warming and further induce a northward shift of the Inter-Tropical Convergence Zone (ITCZ). In this study, we use experimental ensemble climate simulations to show that Antarctic meltwater forcing has distinct regional climate impacts over the globe, leading in particular to regional warming in East Asia. It is suggested that Antarctic meltwater forcing leads to a negative precipitation anomaly in the Western North Pacific (WNP) via cooling in the tropics and the northward shift of the ITCZ. This suppressed convection in WNP induces an anticyclonic flow over the North Pacific, which leads to regional warming in East Asia. This hypothesis is supported by analyses of inter-ensemble spread and long-term control simulations.

1	Impact of Antarctic meltwater forcing on East Asian climate under greenhouse
2	warming
3	Ji-Hoon Oh <sup>1</sup> , Wonsun Park <sup>2</sup> , Hyung-Gyu Lim <sup>1,3</sup> ,
4	Kyung Min Noh <sup>1</sup> , Emilia Kyung Jin <sup>4</sup> , Jong-Seong Kug <sup>1*</sup>
5	
6	<sup>1</sup> Division of Environmental Science and Engineering,
7	Pohang University of Science and Technology (POSTECH), Pohang 37673, South Korea
8	<sup>2</sup> GEOMAR Helmholtz Centre for Ocean Research Kiel, 24105 Kiel, Germany
9	<sup>3</sup> Princeton University/Atmosphere and Oceanic Sciences Program, Princeton, NJ 08540,
10	USA
11	<sup>4</sup> Korea Polar Research Institute (KOPRI), Incheon 21990, South Korea
12	
13	*Corresponding author. Email: jskug@postech.ac.kr

14	Key points
15	• Antarctic meltwater forcing induces an overall global cooling but regional
16	warming in East Asia.
17	• Antarctic meltwater forcing can shift the Inter-Tropical Convergence Zone
18	northward and suppress convection over the Western North Pacific.
19	• Suppressed convection in the Western North Pacific is responsible for the
20	regional warming of East Asia via atmospheric teleconnection.
21	
22	Abstract
23	In recent decades, Antarctic ice-sheet/shelf melting has been accelerated, releasing
24	freshwater into the Southern Ocean. It has been suggested that the meltwater flux could
25	lead to cooling in the Southern Hemisphere, which would retard global warming and
26	further induce a northward shift of the Inter-Tropical Convergence Zone (ITCZ). In this
27	study, we use experimental ensemble climate simulations to show that Antarctic meltwater
28	forcing has distinct regional climate impacts over the globe, leading in particular to
29	regional warming in East Asia. It is suggested that Antarctic meltwater forcing leads to a
30	negative precipitation anomaly in the Western North Pacific (WNP) via cooling in the
31	tropics and the northward shift of the ITCZ. This suppressed convection in WNP induces
32	an anticyclonic flow over the North Pacific, which leads to regional warming in East Asia.
33	This hypothesis is supported by analyses of inter-ensemble spread and long-term control
34	simulations.

#### **Plain Language Summary**

36 In recent decades, greenhouse warming has accelerated the melting of Antarctic glaciers, 37 which discharges freshwater into the Southern Ocean and therefore reduces the surface 38 density. Surface freshening in the Southern Ocean induces cooling and sea-ice expansion 39 on the surface, such that it could delay global warming and further lead to a northward shift 40 of the Inter-Tropical Convergence Zone (ITCZ). Here, we examine the distinct regional 41 impacts of Antarctic meltwater forcing over the globe by analyzing experimental 42 simulations with and without meltwater forcing. For example, the Antarctic meltwater forcing induces a global cooling, but leads to regional warming in East Asia. We find that 43 44 Antarctic meltwater forcing leads to reduced convection in the Western North Pacific 45 (WNP) due to the northward shift of the ITCZ and an overall cooling in the tropics. This 46 circulation change in WNP induces regional warming in East Asia via atmospheric 47 teleconnection.

#### 48 **1. Introduction**

49 Observational evidence has revealed that Antarctic ice-sheet/shelf melting has 50 been accelerating in recent years, and this has resulted in freshwater discharge into the 51 Southern Ocean (Paolo et al., 2015; Wouters et al., 2015; Konrad et al., 2018; Shepherd et 52 al., 2018; Rignot et al., 2019). In a future warmer world, freshwater release from the 53 Antarctic continent may further accelerate (Fogwill et al., 2015; DeConto & Pollard, 2016; 54 Hansen et al., 2016). Nevertheless, the effects of meltwater due to the mass loss from 55 Antarctic ice have not been reflected in future climate projections (Kirtman et al., 2013; 56 Colins et al., 2013) in the Coupled Model Intercomparison Project Phase 5 (CMIP5) and 57 Phase 6 (CMIP6) (Taylor et al., 2012; Eyring et al., 2016).

58 In this context, many previous studies have investigated the impact of meltwater 59 forcing on the climate system by applying an idealized freshwater forcing in climate 60 simulations (Stouffer et al., 2007; Bintanja et al., 2013, 2015; Fogwill et al., 2015; Pauling 61 et al., 2016; Bronselaer et al., 2018; Park & Latif, 2019). Meltwater forcing reduces the 62 surface water density and the oceanic deep convection in the Southern Ocean, which 63 hinders warm Circumpolar Deep Water (CDW) intrusion to the cold surface water; 64 therefore, the intensified stratification in the Southern Ocean leads to cold surface and 65 warm subsurface temperatures around Antarctica, which causes an expansion of the sea-66 ice cover (Bintanja et al., 2013, 2015; Fogwill et al., 2015; Pauling et al., 2016; Park & 67 Latif, 2019). This is accompanied by subsurface warming, leading to an acceleration in 68 basal melting at the bases of the ice shelves (Rignot & Jacobs, 2002; Shepherd et al., 2004; Bintanja et al., 2013, 2015; Obase et al., 2017; Bronselaer et al., 2018). An increase in the 69 70 sea-ice extent delays greenhouse warming in the Southern Ocean via a positive ice-albedo 71 feedback. Moreover, the cooling discrepancy between the Southern Hemisphere and the 72 Northern Hemisphere (Stoker, 1998) alters the atmospheric heat transport, resulting in a 73 northward shift of the Inter-Tropical Convergence Zone (ITCZ) (Zhang & Delworth, 2005; 74 Kang et al., 2008, 2009; Bozbiyik et al., 2011; Cabré et al., 2017; Bronselaer et al., 2018). 75 In fact, satellite observations have recorded a significant expansion in Antarctic 76 sea-ice during the satellite era (Cavalieri & Parkinson, 2008; Comiso & Nishio, 2008) 77 consistent with the cooling trend in the Southern Ocean during recent decades (Zwally et 78 al., 2002; Turner et al., 2009). Conversely, abyssal warming in the Southern Ocean was 79 observed during that same period (Robertson et al., 2011; Purkey & Johnson, 2010, 2012; 80 Fahrbach et al., 2011), possibly due to an enhanced salinity stratification (de Lavergne et 81 al., 2014).

82 Several mechanisms have been suggested to explain the recent observational 83 trends in the Southern Hemisphere. One is the intensification of the Southern Annular 84 Mode (SAM) (Thompson & Wallace, 2000) due to stratospheric ozone depletion, leading 85 to enhanced evaporation from the sea surface in the Southern Ocean (Thompson & 86 Solomon, 2002; Turner et al., 2009). Another candidate is related to surface freshening in 87 response to anthropogenic greenhouse warming, which possibly contributes to the 88 amplification of the global hydrological cycle (de Lavergne et al., 2014) and Antarctic ice-89 shelf melting (Bintanja et al., 2013). In particular, the observational variations are well 90 matched with meltwater-induced climate responses due to Antarctic ice-shelf melting. In 91 addition, long-term internal variability associated with deep convection in the Southern 92 Ocean has been proposed as one of the drivers of the recently observed trend (Latif et al., 93 2013; Zhang et al., 2019).

94 Meanwhile, previous studies investigating the impacts of meltwater have primarily 95 focused on effects in the Southern Hemisphere and do not cover impacts on other areas on 96 the globe, e.g. the Northern Hemisphere and related teleconnection mechanism. Regarding 97 atmospheric teleconnections, tropics are known to play an active role and can regionally 98 affect the extratropical climate in the Northern Hemisphere by modulating tropical 99 convective activities (Hoskins & Karoly, 1981; Lau & Nath, 1994). Many previous studies 100 also have suggested impacts of changes in tropical convection on the East Asian climate 101 via the tropical–extratropical teleconnection (Son et al., 2014; Gong et al., 2015; Kim et 102 al., 2018). Here, we examine the impact of Antarctic meltwater forcing, in particular on 103 East Asia, and suggest a possible mechanism linking Antarctic changes to East Asian 104 climate, where bridge of the tropics is important.

105

#### 106 **2. Data and Methods**

107 To examine the impacts of Antarctic meltwater forcing on global climate, idealized 108 ensemble simulations of the Kiel Climate Model (KCM, Park et al. 2009) were analyzed. 109 KCM is composed of the atmospheric general circulation model (AGCM) ECHAM5 110 coupled with the NEMO ocean/sea-ice GCM. The experimental design of the present study 111 was the same as that of Park & Latif (2019); however, we included additional 12 ensemble 112 members (for a total of 22 ensemble members) to obtain more robust responses to the 113 Antarctic meltwater forcing.

114 The following three different simulations were used in our study to investigate the 115 sensitivity of the climate system to a freshwater forcing in the Antarctic Ocean. The first 116 is a preindustrial control simulation over 2300 years (CTRL) applying a constant CO<sub>2</sub>

117 concentration of 286.2 parts per million (ppm). The other two are global warming 118 simulations with and without Antarctic meltwater forcing. Both simulations were 119 integrated over 200 years with 22 ensemble members. The initial conditions of the 120 individual realizations were taken from CTRL every 100 years and were different for each 121 realization. The first global warming ensemble employed an increasing atmospheric CO<sub>2</sub> concentration at a rate of 1% year<sup>-1</sup> until CO<sub>2</sub>-quadrupling ( $4 \times CO_2$ , 1144.8 ppm); this 122 123 simulation is termed GW. In the other global warming simulation, the increased CO<sub>2</sub> 124 concentration was applied as in GW but with the addition of a freshwater flux to only the 125 Southern Ocean; this simulation is termed GWMW. The total amount of meltwater forcing 126 was 0.1 Sv, and the forcing was exerted proportionally at all of the coastal points of 127 Antarctica describing runoff into the Southern Ocean in the CTRL simulation. Meltwater forcing in GWMW is assumed to be the result of ice-sheet/shelf melting with iceberg 128 129 calving. It is noteworthy that historical and RCP8.5-projected Antarctic meltwater in 130 Deconto & Pollard (2016) reaches 0.1 Sv around the year 2035. Detailed descriptions of 131 experimental design can be obtained in Park & Latif (2019).

Because all the other conditions of the two global warming simulations are identical except for the meltwater forcing, the differences between GW and GWMW imply impact of the meltwater forcing. To test a statistical significance, the bootstrap method and student's t-test were used.

136

137 **3. Results** 

Park and Latif (2019) showed that the major difference between GW and GWMW
is the surface air temperature (SAT) cooling response over the Southern Hemisphere. The

140 fundamental reason for this change is hypothesized to be stabilization in the Southern 141 Ocean due to the freshwater forcing. This causes a cooling tendency at the sea surface and 142 warming in the subsurface, which reduces the oceanic heat loss to the atmosphere and 143 widens the sea-ice cover, eventually leading to a strong positive ice-albedo feedback.

144 To examine the impact of the Antarctic meltwater forcing, we first analyzed the 145 ensemble-mean difference between the GW and GWMW simulations, as shown in Figures 146 1 and 2. Figure 1a shows the evolution of the Southern Hemisphere sea-ice area and the global-mean SAT. As pointed out by previous studies, Antarctic meltwater forcing isolates 147 148 the warm CDW from the surface, resulting in an increase in the sea-ice concentration (SIC) 149 in the Southern Ocean. In the presence of a strong positive albedo feedback, the 150 temperature and sea-ice responses are enhanced. Interestingly, the sea-ice response 151 gradually weakens even though the meltwater forcing is constant. In a linear framework, 152 this is possibly due to ocean adjustments caused by subsurface warming related to the 153 limitation of the deep heat reservoir; a similar mechanism has been suggested in previous 154 studies (Martin et al., 2013; Zhang & Delworth, 2016; Zhang et al., 2017). SAT around the 155 Antarctic region significantly decreases due to the negative downward sensible heat flux 156 and the ice-albedo feedback as a consequence of both cold sea surface temperature (SST) 157 and additional sea-ice formation. Subsequently, the cooling in the Antarctic area plays a 158 role in decreasing the global temperature primarily via atmospheric heat transport, even in 159 the Arctic (Figure 1b).



161 Figure 1. (a) Time series of SAT and sea-ice area in the ensemble-mean difference 162 (GWMW10-GW) smoothed by the 21-year running mean. Global-mean removed SAT in 163 East Asia (blue box in panel (c)) (red), global-mean SAT (blue), and sea-ice area in the 164 Southern Hemisphere (navy blue). The solid lines show the ensemble-mean difference, and the shading shows the 99% uncertainty in the mean. The green box, which indicates the 165 model year period from 22 to 71, indicates the maximum temperature in East Asia. (b) 166 167 Time series of the zonal-mean SAT in the ensemble-mean, smoothed in the same way as in panel (a). Ensemble-mean difference of the (c) SAT (global-mean removed) and (d) sea-168 169 ice concentration, both averaged over the 22–71-year period (green box in panel (a)). The regions denoted by colors indicate where the responses are significant at the 99% 170 171 confidence level.



**Figure 2.** Ensemble-mean difference of the (**a**) precipitation, (**b**) SST ( $5^{\circ}$  S– $5^{\circ}$  N), and (**c**) stream function at 300 hPa (zonal-mean removed) averaged over the 22–71-year period (green box in Figure 1a). The regions denoted by black dots indicate where the responses are significant at the 99% confidence level, and the light-blue shading indicates the 99% uncertainty in the mean. The rightmost red line in panel (a) shows the zonal-mean within  $60^{\circ}$  E– $60^{\circ}$  W.

- 179 180
- The surface cooling is weaker in the mid-latitudes of the Northern Hemisphere than
- 181 that in Southern Hemisphere. To examine the regional dependency of the SAT changes,
- 182 Figure 1c shows the temperature responses to Antarctic meltwater forcing after the global-

183 mean SAT is removed. An interhemispheric contrast is clear, showing cooling in the 184 Antarctic region and relative warming (actually weak cooling) in the mid-latitudes of the 185 Northern Hemisphere. In particular, the warming is most prominent in East Asia and the 186 western part of the North Pacific. This suggests that the temperature response in East Asia 187 to the Antarctic meltwater forcing is determined by a competition between the global 188 cooling effect and a regional warming effect due to the direct and indirect effects of 189 meltwater forcing, respectively. Here, we focus on how Antarctic meltwater forcing leads 190 to the regional warming in East Asia.

The regional warming over East Asia gradually increases from the starting point of the meltwater forcing and has a maximum value at the model years 22–71 (Figure 1a, green box), suggesting that the regional warming response in East Asia has an approximately 21year delay. During this period, the increase in SIC appears in most parts of the Southern Ocean and is strongest in the Weddell Sea, indicating non-uniform sea-ice responses to the meltwater forcing (Figure 1d).

197 To understand how Antarctic meltwater forcing leads to regional warming in East 198 Asia, Figure 2 shows the precipitation, equatorial SST, and stream function at 300 hPa 199 anomalies associated with the meltwater forcing for the same period as in Figure 1c. It is 200 evident that precipitation tends to decrease in the southern tropics and increase in the 201 northern tropics, suggesting a northward shift of ITCZ (Figure 2a). Due to the overall 202 cooling tendency in the tropics, the response in the southern tropics is more distinctive than 203 that in the northern tropics. The Antarctic meltwater forcing leads to negative temperature 204 anomalies in the Southern Hemisphere compared to those in the Northern Hemisphere, 205 which implies an interhemispheric temperature contrast (Stocker, 1998). In this case, the anomalous interhemispheric atmospheric heat transport is toward the Southern Hemisphere,
which possibly induces the northward shift of the ITCZ (Kang et al., 2008, 2009; Bozbiyik
et al., 2011; Cabré et al., 2017; Bronselaer et al., 2018).

209 In addition to the northward shift of ITCZ, it is also clear that the precipitation is 210 significantly reduced over the Western North Pacific (WNP) (Figure 2a). The decrease in 211 the WNP precipitation might be due to the northward shift of ITCZ because the 212 climatological ITCZ is located in WNP. Moreover, the decreased WNP precipitation might 213 be connected to the zonally asymmetric cooling in the tropical Pacific. As shown in Figure 214 2b, this cooling is more significant in the western Pacific than in the eastern Pacific, which 215 indicates a weakened zonal temperature gradient in the equatorial Pacific, reminiscent of 216 an El Niño-like cooling pattern. Such SST variations can induce suppressed atmospheric 217 convection in WNP and activate atmospheric convection in the eastern Pacific.

218 In a Gill-type response to the suppressed convection in WNP, there is an 219 anticyclonic flow in the lower troposphere and a cyclonic flow in the upper troposphere, 220 as shown in Figure 2c (Gill, 1980; Rui & Wang, 2000). The upper-level cyclonic flow and 221 accompanying convergence lead to Rossby wave energy propagation to the extratropical 222 region (Hoskins & Karoly, 1981), which is responsible for a distinctive anticyclonic 223 circulation over the western side of the North Pacific. This circulation pattern is very 224 similar to the so-called Kuroshio anticyclone, which is a response to the suppressed 225 convection in the WNP during the El Niño phase (Son et al., 2014; Kim et al., 2018). The 226 Kuroshio anticyclone has a barotropic structure; therefore, there is still anticyclonic 227 circulation in the lower troposphere. Eventually, this anomalous anticyclonic flow

- accompanies the warm advection in East Asia, which might be responsible for the regional
- 229 warming in East Asia in response to the Antarctic meltwater forcing.



Figure 3. (a) Scatter diagram of SAT in the Equatorial Pacific (red box in panel (e)) versus the precipitation in WNP (blue box in panel (c)) calculated in the inter-ensemble spread for the same period as in Figure 2. (b) Same as panel (a) but for the precipitation in WNP versus SAT (global-mean removed) in East Asia (blue box in panel (d)). Regression maps of the (c) precipitation, (d) stream function at 300 hPa, and (e) SAT onto SAT (globalmean removed) in East Asia. The regions denoted by black dots indicate where the responses are significant at the 99% confidence level.

- -

So far, from the ensemble-mean result, we hypothesize that the Antarctic freshwater discharge induces suppressed convection in WNP, which leads to regional warming in East Asia. To check this hypothesis, we examined these processes in the inter-ensemble spread. 243 If the hypothesis is correct, it should work not only in the ensemble-mean but also in the 244 inter-ensemble spread. The precipitation anomaly in WNP has a strong positive correlation 245 with SAT in the equatorial Pacific with a correlation coefficient of 0.8 (Figure 3a), which 246 is significant at the 99% confidence level. This suggests that, at the given meltwater forcing, 247 the strength of the suppressed convective response in WNP depends on how fast the 248 equatorial Pacific cools down. Further, Figure 3b shows the relationship between 249 precipitation in WNP and SAT in East Asia. It is apparent that ensemble members with 250 more strongly suppressed convection in WNP are prone to simulating stronger warming in 251 East Asia. The correlation between the two is -0.57, which is significant at the 99% 252 confidence level.

253 To further support this argument, the regression with respect to the East Asian SAT 254 index (averaged SAT over the blue box area in Fig. 3d) was computed from the inter-255 ensemble spread. It is interesting that the precipitation pattern (Figure 3c) associated with 256 the regional East Asian warming in the ensemble spread space is in good agreement with 257 the ensemble-mean responses to the meltwater forcing (Figure 2a). For example, the 258 decreased precipitation in WNP and the increased precipitation in the off-equatorial Pacific 259 of the Northern Hemisphere are significant. This similarity indicates that this precipitation 260 pattern is a key for explaining the regional warming in East Asia. As discussed earlier, the precipitation decrease in WNP leads to Rossby wave energy propagation, which is also 261 262 well captured in the regression pattern in the ensemble spread space (Figure 3d). It is also 263 seen that the East Asian warming is related to cooling in the equatorial Pacific (Figure 3e), which might be responsible for the WNP convective response. As in the ensemble-mean, 264 265 the cooling is more dominant in the western Pacific than in the eastern Pacific. These results in the inter-ensemble spread strongly support our hypothesis concerning how Antarctic
meltwater forcing induces regional warming in East Asia through a bridging role of the
tropical Pacific.

269 In addition to the inter-ensemble spread, we also analyzed the 2300-year long-term 270 control integration to further support our hypothesis. To link the East Asian warming to 271 the variability in the Southern Ocean, the sea-ice in the Southern Ocean was regressed with 272 respect to the East Asian SAT index. For consistency, the East Asian SAT index was 273 calculated after the global-mean temperature was removed after applying a 50-year moving 274 average. Figure 4a shows the 21-year leading regression pattern of sea-ice versus the East 275 Asian SAT index. It is evident that regional warming in East Asia is related to the SIC 276 increase in the Weddell Sea, which is consistent with the result in Figure 1d. Moreover, the 277 correlation coefficient between the global-mean removed SAT in East Asia and the SIC in 278 the Weddell Sea has a maximum value when the SIC leads the SAT by 22 years (not 279 shown). This is consistent with the meltwater-induced result that the regional warming in 280 East Asia has a maximum value 21 years after the initialization of Antarctic meltwater 281 forcing.

282 In the control simulation, the spatial patterns of the precipitation, tropical temperature,

and stream function at 300 hPa against the East Asian SAT index are also similar to those

in Figure 2. That is, the suppressed convection in WNP (Figure 4b) and the anticyclonic

- flow at mid-latitudes are distinctive. It seems that the WNP precipitation is sensitive to
- the sea-ice variability in the Weddell Sea, possibly because WNP is a downstream region
- 287 of the Weddell Sea under background westerlies in the Southern Hemisphere. In the case
- 288 of the non-convective phase in the Weddell Sea, a significant precipitation decreases in

WNP also appeared in the study of Latif et al. (2013) and Cabré et al. (2017), which
investigated the global impacts of Southern Ocean internal variability driven by deep
convection changes in the Weddell Sea. These results from the control simulation
indicate that Antarctic sea-ice variability can influence the East Asian temperature via the
convective response in WNP, strongly supporting our hypothesis for the response of East
Asian temperatures to Antarctic meltwater forcing.

- 295
- 296 4. Summary and Discussion

297 We investigated the global teleconnection and associated regional impacts over 298 East Asia due to meltwater forcing in the Southern Ocean under greenhouse warming 299 with a series of climate simulations. In response to the meltwater forcing, surface cooling, 300 abyssal warming in the Southern Ocean, and a northward shift of ITCZ were appeared in 301 this study, as in previous studies (Park & Latif, 2019; Bronselaer et al., 2018). Despite the 302 overall surface cooling trend, the response in East Asia shows slight warming rather than 303 cooling, implying that another process is at work besides the global cooling. This 304 regional warming in East Asia is related to the suppressed convection in WNP, which is 305 caused by both zonally asymmetric cooling in the tropical Pacific and the northward shift 306 of ITCZ. The suppressed convection in WNP induces the Kuroshio anticyclone in the 307 western part of the North Pacific (Son et al., 2014; Kim et al., 2018), which ultimately 308 accompanies warm advection in East Asia. It was also shown that this meltwater-induced 309 regional warming occurs in East Asia with a time lag of approximately 20 years. A 310 statistical analysis of the inter-ensemble spread strongly supports the proposed 311 mechanism for the regional warming in East Asia.



**Figure 4.** Regression maps of the (**a**) SIC, (**b**) precipitation, (**c**) SST, and (**d**) stream function at 300 hPa onto the SAT (global-mean removed) in East Asia. Values are calculated in the CTRL simulation but for the regression results based on the 50-year running mean. In panel (a) SIC leads SAT by 21 years, whereas panels (b), (c), and (d) simultaneous with SAT. The regions denoted by colors indicate where the responses are significant at the 99% confidence level.

320 .The teleconnection mechanism from the Southern Ocean to East Asia can be also 321 realized in natural mode of variability. Latif et al. (2013) reported that the internal 322 centennial variability originating from the Southern Ocean may have affected the recent 323 decadal trends observed in the Southern Hemisphere, such as overall cooling and Antarctic 324 sea-ice expansion. They also showed anomalous warming in the western part of the North 325 Pacific and a weakening of the Aleutian low in the case of the cold phase in the Southern 326 Ocean owing to internal variability. In addition, they argued that the observed weakening 327 of the Aleutian low in recent decades might be related to internal variability originating in the Southern Ocean. Their arguments correspond well with the meltwater-induced responses found in this study. Cooling over the Southern Ocean in their study occurred in response to the shutdown of deep convection due to the internal variability, while cooling in our study is due to meltwater forcing. Even though the cooling sources are different, the resulting teleconnection and remote influence on East Asia may share a similar mechanism.

333 For example, the patterns simulated in the control simulation in Figure 4 result from 334 internal variability in the model because there is no external forcing in that case. This means 335 that the Antarctic-to-East Asia connection is naturally an intrinsic mode in a coupled 336 climate system. In fact, many modeling studies have suggested that the long-term internal 337 variability of the SST and sea-ice extent in the Southern Ocean possibly originate from 338 changes in the deep convection in the Weddell Sea (Martin et al., 2013; Latif et al., 2013, 339 2017; Zunz et al., 2013; Wang & Dommenget, 2016). Our results under their conclusions 340 suggest that Antarctic meltwater forcing possibly triggers this internally intrinsic process, 341 resulting in East Asian regional warming.

342 According to recent studies investigating the impact of Antarctic meltwater on 343 climate, meltwater could induce global cooling and Antarctic sea-ice expansion in the 344 future climate (Bronselaer et al., 2018; Park & Latif, 2019). It is conjectured that the global 345 warming expectation during the 21st century, projected without the impact of Antarctic 346 meltwater, could be delayed by a decade due to meltwater input to the Southern Ocean. 347 However, our results suggest that Antarctic meltwater possibly induces regional warming 348 in East Asia, contrary to the global cooling effect, emphasizing the variety of climate 349 responses associated with Antarctic meltwater in a regional context.

350 Our study includes some limitations. First, our results were based on only one 351 particular climate model, so that they could be model dependent. Nevertheless, Bronselaer 352 et al. (2018), examined the impact of meltwater on the climate system using the 353 Geophysical Fluid Dynamics Laboratory's Earth system model version 2M (GFDL 354 ESM2M), and showed similar responses to our results even though they did not emphasize 355 the regional impact. If more models are utilized, more robust features can be derived. 356 Second, the added meltwater forcing is idealized in this experimental design. The 357 magnitude of the meltwater forcing applied to Antarctica was constant at 0.1 Sv and did 358 not change from the beginning of the model. In addition, the meltwater was only introduced 359 to the surface layer even though some meltwater may be discharged at depth due to basal 360 melting. However, Pauling et al. (2016) reported that this simplification does not have a 361 critical impact on sea-ice and surface temperature variations. Nonetheless, the impact of 362 the meltwater on East Asia and the related possible mechanism is firstly addressed here, 363 which can be further investigated by an approach with a multi-model intercomparison and 364 advanced coupled climate-land ice model.

## 365 Acknowledgments

- 366 This work was sponsored by a research grant from the Korean Ministry of Oceans and
- 367 Fisheries (KIMST20190361; PM19020). W.P. received funding from the European
- 368 Union's Horizon 2020 research and innovation program under grant agreement N°821001
- 369 (SO-CHIC). J.-S. Kug is partly supported by the National Research Foundation of Korea
- 370 (NRF-2017R1A2B3011511). The model data, used in this study, will be available on the
- 371 server, <u>https://data.geomar.de</u>. We did not use any new data in this study.

# **References**

373	Bintanja, R., Van Oldenborgh, G., Drijfhout, S., Wouters, B., & Katsman, C. (2013).
374	Important role for ocean warming and increased ice-shelf melt in Antarctic sea-ice
375	expansion. Nature Geoscience, 6(5), 376-379. https://doi.org/10.1038/ngeo1767
376	Bintanja, R., Van Oldenborgh, G., & Katsman, C. (2015). The effect of increased fresh
377	water from Antarctic ice shelves on future trends in Antarctic sea ice. Annals of
378	Glaciology, 56(69), 120-126. https://doi.org/10.3189/2015AoG69A001
379	Bozbiyik, A., Steinacher, M., Joos, F., Stocker, T., & Menviel, L. (2011). Fingerprints of
380	changes in the terrestrial carbon cycle in response to large reorganizations in ocean
381	circulation. Climate of the Past, 7(1), 319-338. https://doi.org/10.5194/cp-7-319-
382	2011
383	Bronselaer, B., Winton, M., Griffies, S. M., Hurlin, W. J., Rodgers, K. B., Sergienko, O.
384	V., et al. (2018). Change in future climate due to Antarctic meltwater. Nature,
385	564(7734), 53-58. https://doi.org/10.1038/s41586-018-0712-z
386	Cabré, A., Marinov, I., & Gnanadesikan, A. (2017). Global atmospheric teleconnections
387	and multidecadal climate oscillations driven by Southern Ocean convection.
388	Journal of Climate, 30(20), 8107-8126. https://doi.org/10.1175/JCLI-D-16-0741.1
389	Cavalieri, D., & Parkinson, C. (2008). Antarctic sea ice variability and trends, 1979–2006.
390	Journal of Geophysical Research: Oceans, 113(C7), C07004.
391	https://doi.org/10.1029/2007JC004564
392	Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., et al.
393	(2013). Long-term Climate Change: Projections, Commitments and Irreversibility.
394	In: Climate Change 2013: The Physical Science Basis. Contribution of Working

- Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate
  Change [Stocker, T.F., D. Qin, G.-K. Plattner, M. Tignor, S.K. Allen, J. Boschung,
- 397 A. Nauels, Y. Xia, V. Bex and P.M. Midgley (eds.)]. Cambridge University Press,
- 398 Cambridge, United Kingdom and New York, NY, USA.
- Comiso, J. C., & Nishio, F. (2008). Trends in the sea ice cover using enhanced and
  compatible AMSR-E, SSM/I, and SMMR data. *Journal of Geophysical Research: Oceans, 113*(C2), C02S07. https://doi.org/10.1029/2007JC004257
- 402 De Lavergne, C., Palter, J. B., Galbraith, E. D., Bernardello, R., & Marinov, I. (2014).
- 403 Cessation of deep convection in the open Southern Ocean under anthropogenic
  404 climate change. *Nature Climate Change*, 4(4), 278-282.
  405 https://doi.org/10.1038/nclimate2132
- 406 DeConto, R. M., & Pollard, D. (2016). Contribution of Antarctica to past and future sea407 level rise. *Nature*, *531*(7596), 591-597. https://doi.org/10.1038/nature17145
- 408 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K.
- 409 E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6
- 410 (CMIP6) experimental design and organization. Geoscientific Model Development,
- 411 9(5), 1937-1958. https://doi.org/10.5194/gmd-9-1937-2016
- 412 Fahrbach, E., Hoppema, M., Rohardt, G., Boebel, O., Klatt, O., & Wisotzki, A. (2011).
  413 Warming of deep and abyssal water masses along the Greenwich meridian on
  414 decadal time scales: The Weddell gyre as a heat buffer. *Deep Sea Research Part II:*415 *Topical Studies in Oceanography*, 58(25-26), 2509-2523.
  416 https://doi.org/10.1016/j.dsr2.2011.06.007

- 417 Fogwill, C., Phipps, S., Turney, C., & Golledge, N. (2015). Sensitivity of the Southern
- 418 Ocean to enhanced regional Antarctic ice sheet meltwater input. *Earth's Future*,
  419 3(10), 317-329. https://doi.org/10.1002/2015EF000306
- Gill, A. E. (1980). Some simple solutions for heat-induced tropical circulation. *Quarterly Journal of the Royal Meteorological Society*, *106*(449), 447-462.
  https://doi.org/10.1002/qj.49710644905
- Gong, H., Wang, L., Chen, W., Nath, D., Huang, G., & Tao, W. (2015). Diverse influences
  of ENSO on the East Asian–western Pacific winter climate tied to different ENSO
  properties in CMIP5 models. *Journal of Climate*, 28(6), 2187-2202.
  https://doi.org/10.1175/JCLI-D-14-00405.1
- Hansen, J., Sato, M., Hearty, P., Ruedy, R., Kelley, M., Masson-Delmotte, V., et al. (2016).
  Ice melt, sea level rise and superstorms: evidence from paleoclimate data, climate
  modeling, and modern observations that 2 °C global warming could be dangerous. *Atmospheric Chemistry and Physics, 16*(6), 3761-3812.

431 https://doi.org/10.5194/acp-16-3761-2016

- Hoskins, B. J., & Karoly, D. J. (1981). The steady linear response of a spherical atmosphere
  to thermal and orographic forcing. *Journal of the Atmospheric Sciences*, *38*(6),
  1179-1196. https://doi.org/10.1175/1520-
- 435 0469(1981)038%3C1179:TSLROA%3E2.0.CO;2
- Kang, S. M., Frierson, D. M., & Held, I. M. (2009). The tropical response to extratropical
  thermal forcing in an idealized GCM: The importance of radiative feedbacks and
  convective parameterization. *Journal of the Atmospheric Sciences*, *66*(9), 28122827. https://doi.org/10.1175/2009JAS2924.1

440	Kang, S. M., Held, I. M., Frierson, D. M., & Zhao, M. (2008). The response of the ITCZ
441	to extratropical thermal forcing: Idealized slab-ocean experiments with a GCM.
442	Journal of Climate, 21(14), 3521-3532. https://doi.org/10.1175/2007JCLI2146.1

- Kim, S., & Kug, J. S. (2018). What controls ENSO teleconnection to East Asia? Role of
  western North Pacific precipitation in ENSO teleconnection to East Asia. *Journal*of *Geophysical Research: Atmospheres, 123*(18), 10,406-410,422.
  https://doi.org/10.1029/2018JD028935
- Kirtman, B., Power, S. B., Adedoyin, J. A., Boer, G. J., Bojariu, R., Camilloni, I., et al.
  (2013). Near-term Climate Change: Projections and Predictability. In: Climate
  Change 2013: The Physical Science Basis. Contribution of Working Group I to the
  Fifth Assessment Report of the Intergovernmental Panel on Climate Change
  [Stocker, T.F., D. Qin, G.-K. Plattner, M. Tignor, S.K. Allen, J. Boschung, A.
  Nauels, Y. Xia, V. Bex and P.M. Midgley (eds.)]. Cambridge University Press,
  Cambridge, United Kingdom and New York, NY, USA.
- 454 Konrad, H., Shepherd, A., Gilbert, L., Hogg, A. E., McMillan, M., Muir, A., & Slater, T.
- 455 (2018). Net retreat of Antarctic glacier grounding lines. *Nature Geoscience*, 11(4),
  456 258-262. https://doi.org/10.1038/s41561-018-0082-z
- Latif, M., Martin, T., & Park, W. (2013). Southern Ocean sector centennial climate
  variability and recent decadal trends. *Journal of Climate*, 26(19), 7767-7782.
  https://doi.org/10.1175/JCLI-D-12-00281.1
- 460 Latif, M., Martin, T., Reintges, A., & Park, W. (2017). Southern Ocean decadal variability
- 461 and predictability. *Current Climate Change Reports*, *3*(3), 163-173.
  462 https://doi.org/10.1007/s40641-017-0068-8

- Lau, N.-C., & Nath, M. J. (1994). A modeling study of the relative roles of tropical and
  extratropical SST anomalies in the variability of the global atmosphere-ocean
  system. *Journal of Climate*, 7(8), 1184-1207. https://doi.org/10.1175/15200442(1994)007%3C1184:AMSOTR%3E2.0.CO;2
- Martin, T., Park, W., & Latif, M. (2013). Multi-centennial variability controlled by
  Southern Ocean convection in the Kiel Climate Model. *Climate Dynamics*, 40(7),
  2005-2022. https://doi.org/10.1007/s00382-012-1586-7
- Obase, T., Abe-Ouchi, A., Kusahara, K., Hasumi, H., & Ohgaito, R. (2017). Responses of
  basal melting of Antarctic ice shelves to the climatic forcing of the Last Glacial
  Maximum and CO2 doubling. *Journal of Climate*, *30*(10), 3473-3497.
  https://doi.org/10.1175/JCLI-D-15-0908.1
- 474 Paolo, F. S., Fricker, H. A., & Padman, L. (2015). Volume loss from Antarctic ice shelves
  475 is accelerating. *Science*, 348(6232), 327-331.
  476 https://doi.org/10.1126/science.aaa0940
- 477 Park, W., Keenlyside, N., Latif, M., Ströh, A., Redler, R., Roeckner, E., & Madec, G.
  478 (2009). Tropical Pacific climate and its response to global warming in the Kiel
  479 Climate Model. *Journal of Climate, 22*(1), 71-92.
  480 https://doi.org/10.1175/2008JCLI2261.1
- 481 Park, W., & Latif, M. (2019). Ensemble global warming simulations with idealized
  482 Antarctic meltwater input. *Climate Dynamics*, 52(5), 3223-3239.
  483 https://doi.org/10.1007/s00382-018-4319-8
- Pauling, A. G., Bitz, C. M., Smith, I. J., & Langhorne, P. J. (2016). The response of the
  Southern Ocean and Antarctic sea ice to freshwater from ice shelves in an Earth

- 486 system model. *Journal of Climate*, 29(5), 1655-1672. https://doi.org/10.1175/JCLI-
- 487 D-15-0501.1
- Purkey, S. G., & Johnson, G. C. (2010). Warming of global abyssal and deep Southern
  Ocean waters between the 1990s and 2000s: Contributions to global heat and sea
  level rise budgets. *Journal of Climate*, 23(23), 6336-6351.
  https://doi.org/10.1175/2010JCLI3682.1
- 492 Purkey, S. G., & Johnson, G. C. (2012). Global contraction of Antarctic Bottom Water
  493 between the 1980s and 2000s. *Journal of Climate*, 25(17), 5830-5844.
  494 https://doi.org/10.1175/JCLI-D-11-00612.1
- 495 Rignot, E., & Jacobs, S. S. (2002). Rapid bottom melting widespread near Antarctic ice
  496 sheet grounding lines. *Science*, 296(5575), 2020-2023.
  497 https://doi.org/10.1126/science.1070942
- Rignot, E., Mouginot, J., Scheuchl, B., van den Broeke, M., van Wessem, M. J., &
  Morlighem, M. (2019). Four decades of Antarctic Ice Sheet mass balance from
  1979–2017. *Proceedings of the National Academy of Sciences, 116*(4), 1095-1103.
- 501 https://doi.org/10.1073/pnas.1812883116
- Robertson, R., Visbeck, M., Gordon, A. L., & Fahrbach, E. (2002). Long-term temperature
  trends in the deep waters of the Weddell Sea. *Deep Sea Research Part II: Topical Studies in Oceanography*, 49(21), 4791-4806. https://doi.org/10.1016/S09670645(02)00159-5
- Rui, H., & Wang, B. (1990). Development characteristics and dynamic structure of tropical
  intraseasonal convection anomalies. *Journal of the Atmospheric Sciences*, 47(3),

https://doi.org/10.1175/1520-

### 509 0469(1990)047%3C0357:DCADSO%3E2.0.CO;2

357-379.

- 510 Shepherd, A., Ivins, E., Rignot, E., Smith, B., van den Broeke, M., Velicogna, I., et al.
- 511 (2018). Mass balance of the Antarctic Ice Sheet from 1992 to 2017. *Nature*,
  512 558(7709), 219-222. https://doi.org/10.1038/s41586-018-0179-y
- 513 Shepherd, A., Wingham, D., & Rignot, E. (2004). Warm ocean is eroding West Antarctic
  514 ice sheet. *Geophysical Research Letters*, 31(23).
  515 https://doi.org/10.1029/2004GL021106
- Son, H.-Y., Park, J.-Y., Kug, J.-S., Yoo, J., & Kim, C.-H. (2014). Winter precipitation
  variability over Korean Peninsula associated with ENSO. *Climate Dynamics*,
  42(11), 3171-3186. https://doi.org/10.1007/s00382-013-2008-1
- 519 Stocker, T. F. (1998). The seesaw effect. *Science*, 282(5386), 61-62.
  520 https://doi.org/10.1126/science.282.5386.61
- Stouffer, R. J., Seidov, D., & Haupt, B. J. (2007). Climate response to external sources of
  freshwater: North Atlantic versus the Southern Ocean. *Journal of Climate*, 20(3),
  436-448. https://doi.org/10.1175/JCLI4015.1
- Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the
  experiment design. *Bulletin of the American Meteorological Society*, *93*(4), 485498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Thompson, D. W., & Solomon, S. (2002). Interpretation of recent Southern Hemisphere
  climate change. *Science*, 296(5569), 895-899.
  https://doi.org/10.1126/science.1069270

530	Thompson, D. W., & Wallace, J. M. (2000). Annular modes in the extratropical circulation.
531	Part I: Month-to-month variability. Journal of Climate, 13(5), 1000-1016.
532	https://doi.org/10.1175/1520-0442(2000)013%3C1000:AMITEC%3E2.0.CO;2
533	Turner, J., Comiso, J. C., Marshall, G. J., Lachlan-Cope, T. A., Bracegirdle, T., Maksym,
534	T., et al. (2009). Non-annular atmospheric circulation change induced by
535	stratospheric ozone depletion and its role in the recent increase of Antarctic sea ice
536	extent. Geophysical Research Letters, 36(8), L08502.
537	https://doi.org/10.1029/2009GL037524
538	Wang, G., & Dommenget, D. (2016). The leading modes of decadal SST variability in the
539	Southern Ocean in CMIP5 simulations. Climate Dynamics, 47(5), 1775-1792.
540	https://doi.org/10.1007/s00382-015-2932-3
541	Wouters, B., Martin-Español, A., Helm, V., Flament, T., van Wessem, J. M., Ligtenberg,
542	S. R., et al. (2015). Dynamic thinning of glaciers on the Southern Antarctic
543	Peninsula. Science, 348(6237), 899-903. https://doi.org/10.1126/science.aaa5727
544	Zhang, L., & Delworth, T. L. (2016). Impact of the Antarctic bottom water formation on
545	the Weddell Gyre and its northward propagation characteristics in GFDL CM2. 1
546	model. Journal of Geophysical Research: Oceans, 121(8), 5825-5846.
547	https://doi.org/10.1002/2016JC011790
548	Zhang, L., Delworth, T. L., Cooke, W., & Yang, X. (2019). Natural variability of Southern
549	Ocean convection as a driver of observed climate trends. Nature Climate Change,

550 9(1), 59-65. https://doi.org/10.1038/s41558-018-0350-3

551	Zhang, L., Delworth, T. L., & Jia, L. (2017). Diagnosis of decadal predictability of
552	Southern Ocean sea surface temperature in the GFDL CM2. 1 model. Journal of
553	Climate, 30(16), 6309-6328. https://doi.org/10.1175/JCLI-D-16-0537.1

- Zhang, R., & Delworth, T. L. (2005). Simulated tropical response to a substantial
  weakening of the Atlantic thermohaline circulation. *Journal of Climate*, 18(12),
  1853-1860. https://doi.org/10.1175/JCLI3460.1
- 557 Zunz, V., Goosse, H., & Massonnet, F. (2013). How does internal variability influence the
- ability of CMIP5 models to reproduce the recent trend in Southern Ocean sea ice
  extent? *The Cryosphere*, 7(2), 451-468. https://doi.org/10.5194/tc-7-451-2013
- 560 Zwally, H. J., Comiso, J. C., Parkinson, C. L., Cavalieri, D. J., & Gloersen, P. (2002).
- 561 Variability of Antarctic sea ice 1979–1998. *Journal of Geophysical Research:*
- 562 *Oceans*, 107(C5), 9-1-9-19. https://doi.org/10.1029/2000JC000733